Author's response to reviews

Title: A randomised controlled trial of a lifestyle intervention for colorectal cancer survivors (CanChange): study protocol.

Authors:

Anna L Hawkes (annahawkes@cancerqld.org.au)
Kenneth I Pakenham (Anna.hawkes@gmail.com)
Kerry S Courneya (Anna.hawkes@gmail.com)
Sara Gollschewski (Anna.hawkes@gmail.com)
Peter Baade (Anna.hawkes@gmail.com)
Louisa G Gordon (Anna.hawkes@gmail.com)
Brigid M Lynch (Anna.hawkes@gmail.com)
Joanne F Aitken (Anna.hawkes@gmail.com)
Suzanne K Chambers (Anna.hawkes@gmail.com)

Version: 2 Date: 14 August 2009

Author's response to reviews: see over
Revisions to Manuscript ID: 1966731229289722

1. Revision of manuscript title to:
   A randomised controlled trial of a tele-based lifestyle intervention for colorectal cancer survivors (‘CanChange’): study protocol.

2. Revision of author details to:
   Anna L. Hawkes\(^1\)\(^2\)
   Kenneth I. Pakenham\(^3\)
   Kerry S. Courneya\(^4\)\(^{A\_\text{few}}\)
   Sara Gollschewski\(^1\)
   Peter Baade\(^1\)
   Louisa G. Gordon\(^5\)
   Brigid M. Lynch\(^6\)
   Joanne F. Aitken\(^1\)
   Suzanne K. Chambers\(^1\)\(^7\)

   1. Viertel Centre for Research in Cancer Control, Cancer Council Queensland, Brisbane, Australia.
   2. School of Public Health, Queensland University of Technology, Brisbane, Australia
   3. School of Psychology, The University of Queensland, Brisbane, Australia
   4. Faculty of Physical Education and Recreation, University of Alberta, Edmonton, Canada.
   5. Queensland Institute of Medical Research, Brisbane, Australia.
   6. Cancer Prevention Research Centre, The University of Queensland, Brisbane, Australia.
   7. Griffith Institute for Health and Medical Research, Griffith University, Brisbane, Australia.

3. Revision of references to:
   a. Full details have been provided for reference number 11
   b. A year has been provided for reference number 36
   c. Full details have been provided for reference number 38
   d. References number 18, 28-31 have been renumbered as recommended – apologies for the confusion
   e. An additional reference has been included (number 47) which has resulted in renumbering of references

4. Minor revisions including:
   a. The word title has been removed from the title page
   b. Trial registration number has been removed to the end of the abstract
c. Underlining has been removed

5. The manuscript has been checked for typographical errors and changes made in ‘track changes’.

6. Revisions to the Author Contributions section